Author's response to reviews

Title: Design of the Pacemaker REmote Follow-up Evaluation and Review (PREFER) trial to assess the clinical value of the remote pacemaker interrogation in the management of pacemaker patients

Authors:

Jane Chen (janechen@im.wustl.edu)
Bruce L Wilkoff (wilkofb@ccf.org)
Wassim Choucair (choucair@yahoo.com)
Todd J Cohen (tcohen@winthrop.org)
George H Crossley (ghcmd@comcast.net)
W Ben Johnson (wbjohnson@iowaheart.com)
Luc R Mongeon (luc.mongeon@medtronic.com)
Gerald A Serwer (gserwer@umich.edu)
Lou Sherfesee (lou.n.sherfesee@medtronic.com)

Version: 2 Date: 24 March 2008

Author's response to reviews:

Dear Trial Editors:

Please find attached a manuscript entitled: "Design of the Pacemaker REmote Follow-up Evaluation and Review (PREFER) trial to assess the clinical value of the remote pacemaker interrogation in the management of pacemaker patients ". This is a revision of the previously reviewed manuscript: MS: 1747696046180318 which I am submitting on behalf of my co-authors. We have carefully considered all the issues by the reviewers and have made the requested formatting changes to the manuscript.

Thank you for your consideration of this article.

Sincerely,

Jane Chen, M.D. on behalf of the PREFER investigators
Washington University School of Medicine
660 South Euclid Avenue, Campus Box 8086
St. Louis, Missouri 63110
314-454-7877 (office)
314-454-8250 (fax)
janechen@im.wustl.edu